Author's response to reviews

Title: Influenza immunisation, inverse care and homelessness: cross-sectional service-improvement study of eligibility and uptake

Authors:

Alistair Story (Alistairstory@gmail.com)
Robert W Aldridge (rob.aldridge@gmail.com)
Tat Gray (tatgray@gmail.com)
Stan Burridge (info@londonpathway.org.uk)
Andrew C Hayward (a.hayward@ucl.ac.uk)

Version: 2 Date: 31 December 2012

Author's response to reviews:

Description of changes made to BMC Public Health MS: 1620090930859273
Research article: Influenza immunisation, inverse care and homelessness: cross-sectional service-improvement study of eligibility and uptake

Alistair Story, Robert W Aldridge, Tat Gray, Stan Burridge and Andrew C Hayward

1. Abstract formatted according to guidelines with study design included in methods section.
2. All funding statements moved to the acknowledgements section
3. Competing interests section formatted
4. Statement on ethical approval included in methods section. Ethical approval: the research was based on analysis of anonymous client questionnaires collected by an NHS service to inform service delivery. Clients gave verbal consent for collection of data. As such no ethical approval was required.
5. Study questionnaire uploaded